Author’s response to reviews

Title: Clinical course of IgG4-related hypophysitis presenting with focal seizure and relapsing lymphocytic hypophysitis.

Authors:
Kanchana Ngaosuwan (dream.kanchana@gmail.com)
Therdkiat Trongwongsa (therdkiat.g@swu.ac.th)
Shanop Shuangshoti (shanop@gmail.com)

Version: 2 Date: 14 Oct 2015

Author’s response to reviews:

Response to the reviewer’s comments
Re: MS: BEND-D-15-00045R1
Reviewer reports:

The authors were suggested to do extensive proofreading for their manuscript by the reviewer, but they have not completed it in the revised form. references 21, 30, 33 and 37 should be checked.

As suggest by the reviewer, we have done extensive proofreading for our manuscript and highlighting the modification in the yellow color.

To edit the bibliography, references 21 and 30 were manually changed to “DOI” reference style.

Reference 33 was manually created according to author instruction in BMC Endocrine Disorders reference style.

Reference 37 was changed its page number refer to the journal’s hard copy.
All of the reference styles were modified as follows:

Ref 21

Ref 30

Ref 33
Ref 37